Author's response to reviews

Title: Evaluation of high-risk Human papillomaviruses type distribution in cervical cancer in Sichuan province of China

Authors:

En-qi Wu (w1n2q3@yahoo.com.cn)
Guo-nan Zhang (zhanggn@hotmail.com)
Xiang-hui Yu (xianghui@jlu.edu.cn)
Yuan Ren (renyuan811@163.com)
Ying Fan (fanying51@163.com)
Yong-ge Wu (yonggewu@163.com)
Wei Kong (weikong@jlu.edu.cn)
Xiao Zha (zha530909@sina.com)

Version: 7 Date: 20 July 2008

Author's response to reviews:

Dear Editor:

Thank you very much for your core pre-accept e-mail on 17th of this month. According to your asking we have checked our manuscript again carefully and found some typographical mistakes in it, and we have made the changes as follows:

1. Background section- 1st paragraph, line 3 (in the PDF document), the word “women” is replaced by “them”.
2. Result section- 3rd paragraph, line 1, the number “4 of 166” is changed to “4 of 176”, “24 of 167” is changed to “24 of 177”.
3. Discussion section- 1st paragraph, line 2, the number “85.8% (163/190)” is changed to “86.8% (165/190)”.
4. Discussion section- 1st paragraph, line 2, the word “consistent with” is replaced by “close to”.
5. Discussion section- 1st paragraph, page 10, line 5, the number “five” is changed to “six”.
6. Discussion section- 1st paragraph, page 10, line 10, the number “4” is changed to “6”.
8. Table 3- the number “4 (2.1)” is changed to “6 (3.1)”, “27 (14.2)” is changed to “25 (13.2)”, “143 (75.3)” is changed to “145 (76.3)”, “47 (24.7)” is changed to “45 (23.7)”.

Thank you very much for your patient.
With best wishes

Sincerely yours,

Xiao Zha

Department of Molecular Pharmacology and Gynecol Oncology,
Sichuan Tumor Hospital, Chengdu 610041, China